Author's response to reviews

Title: Evaluation of variants in the selectin genes in age-related macular degeneration.

Authors:

Robert F Mullins (robert-mullins@uiowa.edu)
Jessica M Skeie (jessica-skeie@uiowa.edu)
James C Folk (james-Folk@uiowa.edu)
Frances M Solivan-Timpe (frances-solivan@uiowa.edu)
Thomas A Oetting (thomas-oetting@uiowa.edu)
Jian Huang (jian-huang@uiowa.edu)
Kai Wang (kai-wang@uiowa.edu)
Edwin M Stone (edwin-stone@uiowa.edu)
John H Fingert (john-fingert@uiowa.edu)

Version: 7  Date: 19 April 2011

Author's response to reviews: see over
April 19, 2011,

Dear BMC Medical Genetics Editors,

Thank you for your careful editing of our manuscript “Evaluation of variants in the selectin genes in age-related macular degeneration” MS: 1994253859452625.

We have addressed each of the following edit requests:

1) The authors have no competing interests. A section with this statement was added between the Conclusions and the Authors contributions.

We have made the requested changes and look forward to your final review.

Best regards,

John H Fingert, MD PHD
Associate Professor of Ophthalmology and Visual Sciences